CENWP-OP-B                                                                                                                                  13 August 2006
MEMORANDUM FOR CENWP-OP, ATTN: Randy Bailey

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 32 of 2006, which covers the period from 06 through 12 August 2006.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 126.4 to 152.5 kcfs.  Daily average project tailwater elevation ranged from 11.7’ to 13.5’ msl.  Daily average powerhouse forebay elevation ranged from 75.4’ to 76.2 msl.  Daily average river temperature ranged between 70( and 72(F.  Secchi disk measurements averaged 5.0 to 7.0 feet.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B-Branch respectively, are included in the spill pattern.  Daily average spill ranged from 83.5 to 90.3 kcfs.
1.3 Unit Operation.

1.3.1 Unit 0 remains out of service due to governor trouble.

1.3.2 Unit 2 remains out of service for turbine rehab.
1.3.3 Unit 10 remains out of service for turbine rehab.
1.3.4 Unit 14 had a 1% violation from 1100 until 1335 on 07 August while BPA calibrated the T11/12.
1.3.5 Unit 15 remains out of service while waiting for spare parts.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby, as necessary, during nighttime hours to float debris. 
1.4.1 F1 was out of service from 0500 until 1715 on 08 August for brush rigging PMs and to facilitate the ROV fishway inspection.

1.4.2 F2 was out of service from 0500 until 1718 on 08 August for brush rigging PMs and to facilitate the ROV fishway inspection.
1.5 Gatewell drawdowns are measured on Wednesdays; VBS drawdowns are measured daily on modified units.
2 ACTIVITIES

2.1 AWS Closures.  The following AWS valves were closed.
2.2 STS/VBS Inspections.  No STSs/VBSs were inspected this reporting week.
	Unit
	Date
	Run Hours
	Comments

	11-18
	NA
	NA
	NA


Table 1.  STS and VBS inspections.

2.3 Dewatering and Fish Salvage.  No dewatering activities occurred this reporting week.
3 RESEARCH
3.1 CRITFC.  Sampling adult fish in the AFF.  Every other day restriction due to temperatures reaching 70(F
3.2 ERDC TDG.  Sensors remain installed in the tailrace.

3.3 NOAA Fisheries- Gilbreath.  Salt water rearing subyearling chinook at the SMF continues.   
3.4 NOAA Fisheries- Jacobson.   Collecting subyearling chinook for growth and survival work.  

3.5 NOAA Fisheries- McComas.  Acoustic tagging in the SMF concluded in July.

3.6 NOAA Fisheries lamprey.  Sampling continues.
3.7 PSMFC smolt monitoring.  Adopted FPAC sampling protocols when water temps reached 70ºF.  Target sample size was reduced to 100 fish per day and processing every two hours throughout the day.  12 subyearling chinook were collected for NOAA researchers.  
3.8 PSMFC PIT tag.  Monthly testing of the B2CC detectors June through October.
3.9 PSMFC/USDA.  Pikeminnow angling concluded this week.

3.10 USGS lamprey.  Lamprey to be held in the FERL starting the week of August 14th.
3.11 WDFW Fish Counting.  Visual fish counting continues.

4 FISHWAYS 

4.1 Project biologists inspected 06 through 12 August 2006.  
4.2  Fish Passage Plan Observations.
	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR#

	All week
	DSM2 airburst
	Five screens not working.
	Electrical
	See 4.7.2
	06-40863

	8/6
	FG3-26
	Incorrect position for TW
	System in manual
	Operators closed diff.
	

	8/6
	A-branch 
	Reading <1.3’>
	Unknown
	No action taken.
	

	8/7
	PH2
	Units out of FPP priority
	BPA calibration activities
	Back in criteria by 1335.
	

	8/8
	WA Shore fishway
	Water elevations low
	ROV inspection
	Back in FPP criteria by end of the day
	

	8/8-12
	DSM2
	Orifice F2B-S stuck 
	Separated from the shaft.
	Local/closed
	06-50979

	8/8, 8/10-11
	Weir 37
	Reading <1.3’>
	Unknown
	No action taken.
	

	8/8-12
	Weir 38
	Reading <1.3’>
	Unknown
	No action taken.
	

	8/9-12
	DSM2
	11B-S
	Bolts connecting the actuator to the valve are broken.
	Local/open
	06-51106


Table 2.  FPP Out of Criteria items.

4.2.1 PH1 electronic velocity meter is in an eddy.  Due to this placement, the velocity meter reads less than 1.5 fps.  Readings to meet FPP criteria are taken from the south end mechanical meter.
4.2.2 PH2 electronic velocity meter was incorrectly installed in the winter of 04/05.

4.2.3 Cascades Island FG6-12 is closed due to a bent shaft.  Repairs can not be made until winter 06/07.  this diffuser is out of criteria when tailwater exceeds 13’
4.2.4 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Diffuser failure occurred during water up.  Repairs are scheduled for winter 07/08.  For now, the collection channel is operating within FPP criteria.

4.3 Fishway Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  
4.3.1 Due to a Palm Pilot malfunction, temperatures will not be reported this week.  We will include these missing temperatures next week.  Researchers should use the guidelines for sampling at 70(F.
	Date
	WS avg.
	BI avg.
	AFF max
	SMF max

	8/6-12
	NA
	NA
	NA
	NA


Table 3.  Bonneville Fishway temperatures.
4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  
4.5 Adult Fishways.    
4.5.1 Calibration is checked on the weekends.  B-branch tailwater is significantly out of calibration.  

4.5.2 Cascades Island and B-Branch PLCs are in manual.  
4.5.3 Due to temperatures at the AFF reaching 70(F, trapping is limited to every other day.

4.5.4 On 08 August, the WA Shore and Cascades Is. mid-season ROV fishway inspection was completed.  PH1 and B-Branch will be completed on 16 August.
4.5.5 Ladders will return to non-shad passage mode on 16 August.
4.6 Juvenile Fishways.
4.6.1 DSM1 remains out of service until 15 September.

4.6.2 DSM2 airburst screens 1, 4, 5, 11 and 12 do not operate consistently.  The Project is waiting on contractors before repairs may be completed.

4.6.3 Due to temperatures at the SMF reaching 70(F or greater, personnel will process fish more often to reduce holding time.  

5 CONSTRUCTION
5.1 B1 ITS Wall Removal.  Plans are to remove the wall, slope the floor, and install three automated gates.
5.2 PH1 STS/VBS removal and TIES removal.  Six TIES and 23 PH1 STSs/VBSs will be scrapped.

5.3 SMF Full flow PIT Tag Detectors.  Testing ongoing.
6 HAZMAT, SPILLS AND CLEANUP - Nothing to report.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS. 
7.1 A teletype sent on 28 June requests that Bonneville start summer spill at or above 75 KCFS during daytime hours and spill to the 120% gas cap at night until further notice.
7.2 A teletype sent on 11 August requests that when the project is: unable to maintain 30 KCFS generation; spill at or above 75 KCFS during the day; spill to the gas cap at night; AND maintain the forebay soft constraint, then do the following-

7.2.1 Maintaining the forebay soft constraint is least important.

7.2.2 During daytime hours, spill at or above 75 KCFS and generate as needed (30 KCFS minimum).

7.2.3 During nighttime hours, generate 30 KCFS and spill a minimum of 50 KCFS. 

7.2.4  If it is determined the project cannot maintain the 120% TDG spill level through the night then reduce spill as necessary to maintain forebay in the normal operating range but no not fill over the period.  Make best efforts to maintain constant spill across the period.

7.2.5 See Table Bon-6 on page BON-14 in the FPP dated March 2006 for specification of daytime hours.
7.3 A teletype sent on 11 August requests a tailwater restriction of less than 14’ from 1000 until 1600 on 12 August.  This is to allow divers to remove an obstruction in the downstream navlock gate.  

7.4 A teletype sent on 26 July requests a soft forebay constraint of 75.0’ – 76.5’ from 0600 until 1800 Mon- Sat from 07 through 12 August.  This operation is necessary for SDS Lumber to maneuver a barge for rock placement.  Additional teletypes will be issued, as needed, until the work is completed 26 August.
7.5 A teletype sent on 11 August requests a soft forebay constraint of 75.0’ – 76.5’ from 0600 until 1800 Mon- Sat from 14 through 19 August.  This operation is necessary for SDS Lumber to maneuver a barge for rock placement.  Additional teletypes will be issued, as needed, until the work is completed 26 August.
JAMES R. MAHAR, P.E.

Operations Manager
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